Author’s response to reviews

Title: A woman’s worth: An access framework for integrating Emergency Medicine with Maternal Health to reduce the burden of maternal mortality in sub-Saharan Africa

Authors:

Martina Anto-Ocrah (martina_ocrah@urmc.rochester.edu)
Jeremy Cushman (Jeremy_Cushman@urmc.rochester.edu)
Mechelle Sanders (mechelle_sanders@urmc.rochester.edu)
Timothy Dye (tim_dye@urmc.rochester.edu)

Version: 2 Date: 27 Dec 2019

Author’s response to reviews:

Comment 1. Please state in the cover letter whether the image depicted in image 1 are your own or taken from another source.

Response: The legend in the image has read as follows, and explicitly details the source. To our knowledge, the image is not copyrighted and is publicly available at the link below:


Comment 2. Please format your abstract and main text to include the following subheadings: Background, Main Text and Conclusions.
Response 2: abstract and main text have been formatted to include the following subheadings: Background, Main Text and Conclusions.

Comment 3: Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Response 3: the list has been updated and placed after the Conclusions Section as noted.

Comment 4: Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response 4: we have uploaded a clean version of the manuscript, with no tracked changes as requested.